Author’s response to reviews

Title: The role of text messaging intervention in Inner Mongolia among patients with type 2 diabetes mellitus: a randomized controlled trial

Authors:
Xuemei Wang (wangxm_zsu@163.com)
Dan Liu (liudan_summer@163.com)
Maolin Du (duaolin2016@163.com)
Ruiqi Hao (haorqimu@163.com)
Huiqiu Zheng (zhxqqjye1988@163.com)
Chaoli Yan (aliceyzl@126.com)

Version: 2 Date: 05 May 2020

Author’s response to reviews:

Dear editor,

We really appreciate your great effort in reviewing our manuscript and provide us an opportunity to improve the quality of our paper.

According to the editor suggestions, we have carefully revised and improved our manuscript. In addition, we revised it according to the Submission Guidelines. All authors declared that they have no competing interests.

Your comments and suggestions really helped us a lot. We have put great efforts on them. If have any information I can provide, please don’t hesitate to contact me.

Thank you again for your time and patience. Looking forward to hearing from you.

Sincerely yours,
Xuemei Wang,
Maolin Du.

Manuscript ID: MIDM-D-19-00579
Type of manuscript: Article
Title: The role of text messaging intervention in Inner Mongolia among patients with type 2 diabetes mellitus: a randomized controlled trial

FORMATTING CHANGES:
Point: We have also noted that author Maolin Du's initials and contributions are missing from the "authors' contributions" section of the "declarations". Please include this authors' initials with their specific contribution to the study.
Response: Thank you for the recommendations. Based on your suggestions, we have added Maolin Du's initials and contributions in the "authors' contributions" section of the "declarations".